Author's response to reviews

Title: Aspirin for the prevention of cognitive decline in the elderly: rationale and design of a neuro-vascular imaging study (ENVIS-ion)

Authors:

Christopher M Reid (chris.reid@monash.edu)
Elsdon Storey (elsdon.storey@monash.edu)
Tien Y Wong (ophwty@nus.edu.sg)
Robyn Woods (robyn.woods@monash.edu)
Andrew Tonkin (andrew.tonkin@monash.edu)
Jie Jin Wang (jiejin.wang@monash.edu.au)
Anthony Kam (a.kam@alfred.org.au)
Andrew Janke (andrew.janke@cai.uq.edu.au)
Rohan Essex (rohanessex@hotmail.com)
Walter P Abhayaratna (walter.abhayaratna@act.gov.au)
Marc M Budge (marc.budge@anu.edu.au)

Version: 3 Date: 1 February 2012

Author's response to reviews: see over
Dear Editorial Team

RE: MS ID: 9937322446234728
Aspirin for the Prevention of Cognitive Decline in the Elderly: Rationale and Design of a Neuro-Vascular Imaging Study (ENVIS-ion)

Thankyou for your email requesting formatting changes to our manuscript.

The changes that have been made are:

- Abstract structure has been changed to: Background, Methods/Design, Discussion.
- The trial registration has been included in the abstract.
- Author titles and qualifications have been removed
- Statement of Funding has been moved to the Acknowledgements section
- “Introduction” has been changed to “Background” and “Methods” to “Methods/design”.
- The legend has been removed from the figure file and the figure cropped to minimize the white space.

Yours sincerely,

[Signature]

Professor Christopher M. Reid  
ASPREE Coordinating Centre  
School of Public health and Preventive Medicine  
Monash University